Author’s response to reviews

Title: Prevalence of dysmenorrhea and predictors of its pain intensity among Palestinian female university students

Authors:

waleed sweileh (waleedsweileh@yahoo.com)

Heba A.Abu Helwa (hebaabuhelwa17@yahoo.com)

Areen Mitaeb (areen-m@hotmail.com)

Suha Al-Hamshri (s.hamshari@najah.edu)

Version: 2 Date: 09 Jan 2018

Author’s response to reviews:

Dear editor,

thanks for the email and comments.

I did all the required changes pertaining to

1. ethical approval section within the manuscript

2. Competing interest

3. Consent for publication

4. Co-author contribution of SAH

Best Regard